Author's response to reviews

Title: Exploring Diabetes Burden in Mongolia Through a National Knowledge, Attitudes and Practices Survey

Authors:

Alessandro R Demaio (demaio@sund.ku.dk)
Dugee Otgontuya (otgon_phi@hotmail.com)
Maximilian de Courten (maxc@sund.ku.dk)
Ib C Bygbjerg (iby@sund.ku.dk)
Palam Enkhtuya (p_enkhee2001@yahoo.com)
Janchiv Oyunbileg (jobileg@gmail.com)
Dan W Meyrowitsch (dame@sund.ku.dk)

Version: 5 Date: 7 March 2013

Author's response to reviews: see over
RE: MS 9930329279281197, EXPLORING KNOWLEDGE, ATTITUDES AND PRACTICES RELATED TO DIABETES IN MONGOLIA: A NATIONAL POPULATION-BASED SURVEY

Dear Reviewers,

Thank you for your continued support and feedback.

________________________________________

Editor

I have one additional suggested amendment, and the manuscript should only be accepted under the provision that this amendment is conducted: All tables and the figure should give complete information concerning time (when was survey conducted?), place (Mongolia) and person (number of participants, age range, gender).

Thank you. This has been amended.